
Summary of Installation Activities: 3/16/2006 
L1 Calorimeter Trigger Upgrade 

 
 
1. More safety hardware work was done at MCH1. All should be done by tomorrow.  
2. Question was raised about the installation of the BLS cables on the patch panel 

cards, i.e. if the connectors were latching at the location on the cards. I checked 
some of them to conclude that the procedure was OK. I just need to make sure 
that tension on the cables will not pull them out when the panel is closed. The 
clamps help but not always. I installed some of the pleated foil cables. The clamps 
are pain at the …….., if I may say so. 

3. We pulled the cables about a foot from under the floor for panel number 4 on 
M112, one that I could not install yesterday (they were short). 

4. Peter and I had the first pulser run this afternoon. Dean explained the signal, how 
it should look like etc. I only checked part of the panel. I will check the entire 
rack after I install the fourth panel cards and the cables. 

5. Because of all these activities, I did not have chance to install any new cables 
today. I will tomorrow.  

6. At the 9:15 meeting this morning, I believe the consensus was that we should be 
able to start moving the sidewalk equipment to MCH1 on Monday next week 
(3/20).  

7. Linda has the drawings she needed from John F. and, will start the ORC 
procedure tomorrow. 

8. Work on the sidewalk continued. Again, I was away from the sidewalk all day, 
and I think Mike and Ted continued to work on L2 and L3 issues. 

 
Please see the web for details. URL is 
http://d0server1.fnal.gov/users/selcuk/www/default.htm. This information is on 
the trigger online logbook also. 

 
Selcuk Cihangir 

 


